Author’s response to reviews

Title: Detection of Early Lung Cancer Among Military Personnel (DECAMP) Consortium: Study Protocols

Authors:

Ehab Billatos (ebillato@bu.edu)
Fenghai Duan (fduan@stat.brown.edu)
Elizabeth Moses (emoses@bu.edu)
Helga Marques (hmarques@stat.brown.edu)
Irene Mahon (IMahon@acr.org)
Lindsey Dymond (ldymond@acr.org)
Charles Apgar (CAPGAR@acr.org)
Denise Aberle (DAberle@mednet.ucla.edu)
George Washko (gwashko@bwh.harvard.edu)
Avrum Spira (aspira@bu.edu)

Version: 1 Date: 15 Nov 2018

Author’s response to reviews:

I have emailed the ethical proof documentation for each site (IRB approval for all the participating sites). Regarding funding documentation, we do not typically share funding awards documents but I can give you the funding award number. This work was supported by funds from the Department of Defense (W81XWH-11-2-0161), the National Cancer Institute (U01CA196408), and Janssen Research & Development LLC.

As for peer review by the funding body, it was peer-reviewed by the Department of Defense. Specifically is the Lung Cancer Research Program (LCRP) of the DoD.